\begin{table}%tA.1
\caption{\label{table:crossref}Overview of the variable stars treated in Sects.~\ref{sect:B}$-$\ref{sect:other}. The subsequent columns denote our star identification used in this paper, the WEBDA~star number if available, the~$\alpha$ and $\delta$~coordinates of the star, the spectral type taken from SIMBAD, the mean Geneva~$V$, $B_2-V_1$ and $B_2-U$~photometry, the main frequency and its amplitude and the section and figures of this paper where the star is treated. Sections~\ref{sect:Bmulti}$-$\ref{sect:Bnoper} refer to the multi-periodic, mono-periodic and irregular B~stars, Sects.~\ref{sect:AFmulti} and~\ref{sect:AFmono} to the multi- and mono-periodic~A and F~stars and Sect.~\ref{sect:other} to the other variable stars.}
\par
%\centering
\par
\begin{tabular}{cccccrrrrrrcr@{}l}
\hline\hline\noalign{\smallskip}
Star ID & WEBDA & $\alpha$ & $\delta$ & SpT & $V$ &
$B_2-V_1$ & $B_2-U$ & \multicolumn{2}{c}{Frequency} & Ampl. & Section &
\multicolumn{2}{c}{Figures}\\
~ & ~ & (h m s) & ($\degr~'~''$) & ~ & ~ & ~ & ~ & (d$^{-1}$) & ($\mu$Hz) &
(mmag) & ~ & ~\\
\hline
00004 & 2299 & 02 22 08.55 & +57 07 28 & B0.5IV &    9.15 &   0.24 &   0.91 &   
3.145 &   36.40 &    6.3 & \ref{sect:Bmulti} & \ref{OLMFpage}, &
\ref{fig:00004b} \\
00011 & 2444 & 02 22 29.85 & +57 12 28 & ~ &    9.49 &   0.19 &   0.82 &   
4.582 &   53.03 &    5.8 & \ref{sect:Bmulti} & \ref{fig:00011a}, &
\ref{fig:00011b} \\
00012 & 2246 & 02 22 02.76 & +57 08 25 & B2III &    9.94 &   0.16 &   0.81 &   
5.429 &   62.84 &    5.6 & \ref{sect:Bmulti} & \ref{fig:00012a}, &
\ref{fig:00012b} \\
00013 & 2572 & 02 22 48.95 & +57 09 14 & ~ &    9.99 &   0.20 &   0.76 &   
4.415 &   51.10 &    5.2 & \ref{sect:Bmulti} & \ref{fig:00013a}, &
\ref{fig:00013b} \\
00014 & 2488 & 02 22 36.28 & +57 04 54 & B1V &    9.96 &   0.14 &   0.91 &   
6.168 &   71.39 &    6.9 & \ref{sect:Bmulti} & \ref{fig:00014a}, &
\ref{fig:00014b} \\
00024 & 2242 & 02 22 02.45 & +57 09 20 & B1Vnp &   10.93 &   0.23 &   0.65 &   
1.569 &   18.16 &   12.8 & \ref{sect:Bmulti} & \ref{fig:00024a}, &
\ref{fig:00024b} \\
00027 & 2185 & 02 21 55.55 & +57 05 58 & B3V &   10.95 &   0.17 &   0.52 &   
0.625 &    7.23 &    1.2 & \ref{sect:Bmulti} & \ref{fig:00027a}, &
\ref{fig:00027b} \\
00030 & 2114 & 02 21 46.71 & +57 07 27 & B2V &   11.05 &   0.18 &   0.72 &   
0.340 &    3.94 &    7.6 & \ref{sect:Bmulti} & \ref{fig:00030a}, &
\ref{fig:00030b} \\
00041 & 2091 & 02 21 44.46 & +57 10 52 & B3 &   11.62 &   0.28 &   0.48 &   
2.511 &   29.06 &    2.6 & \ref{sect:Bmulti} & \ref{fig:00041a}, &
\ref{fig:00041b} \\
00045 & 2191 & 02 21 56.93 & +57 12 20 & ~ &   11.65 &   0.28 &   0.51 &   
0.328 &    3.80 &    4.0 & \ref{sect:Bmulti} & \ref{fig:00045a}, &
\ref{fig:00045b} \\
00055 & 2094 & 02 21 44.38 & +57 08 54 & B2 &   11.85 &   0.19 &   0.69 &   
0.376 &    4.35 &    4.1 & \ref{sect:Bmulti} & \ref{fig:00055a}, &
\ref{fig:00055b} \\
00073 & 2507 & 02 22 38.85 & +57 11 44 & ~ &   12.49 &   0.20 &   0.47 &   
2.407 &   27.86 &    2.8 & \ref{sect:Bmulti} & \ref{fig:00073a}, &
\ref{fig:00073b} \\
00079 & 2245 & 02 22 02.77 & +57 08 52 & B5 &   12.49 &   0.20 &   0.50 &   
2.670 &   30.91 &    2.3 & \ref{sect:Bmulti} & \ref{fig:00079a}, &
\ref{fig:00079b} \\
00084 & 2110 & 02 21 45.95 & +57 05 01 & ~ &   12.65 &   0.27 &   0.33 &   
2.600 &   30.10 &    1.7 & \ref{sect:Bmulti} & \ref{fig:00084a}, &
\ref{fig:00084b} \\
00088 & 2253 & 02 22 03.48 & +57 07 08 & B7 &   12.66 &   0.23 &   0.35 &   
0.714 &    8.27 &    9.1 & \ref{sect:Bmulti} & \ref{fig:00088a}, &
\ref{fig:00088b} \\
00100 & 2349 & 02 22 14.63 & +57 06 13 & B7 &   12.83 &   0.18 &   0.41 &   
2.699 &   31.24 &    2.3 & \ref{sect:Bmulti} & \ref{fig:00100a}, &
\ref{fig:00100b} \\
00103 & 2429 & 02 22 27.25 & +57 09 07 & ~ &   12.93 &   0.17 &   0.44 &   
1.083 &   12.53 &    3.5 & \ref{sect:Bmulti} & \ref{fig:00103a}, &
\ref{fig:00103b} \\
00115 & 2319 & 02 22 10.62 & +57 06 40 & B8 &   13.11 &   0.19 &   0.31 &   
3.065 &   35.47 &    5.6 & \ref{sect:Bmulti} & \ref{fig:00115a}, &
\ref{fig:00115b} \\
00118 & 2345 & 02 22 13.69 & +57 05 17 & B3V &   13.08 &   0.18 &   0.43 &   
2.000 &   23.15 &    9.2 & \ref{sect:Bmulti} & \ref{fig:00118a}, &
\ref{fig:00118b} \\
00128 & 2267 & 02 22 04.90 & +57 06 01 & B7 &   13.25 &   0.20 &   0.29 &   
2.603 &   30.13 &    3.6 & \ref{sect:Bmulti} & \ref{fig:00128a}, &
\ref{fig:00128b} \\
00138 & 2228 & 02 22 00.35 & +57 04 36 & ~ &   13.57 &   0.32 &  --0.18 &   
3.799 &   43.97 &    1.1 & \ref{sect:Bmulti} & \ref{fig:00138a}, &
\ref{fig:00138b} \\
00145 & 2515 & 02 22 39.65 & +57 04 49 & ~ &   13.48 &   0.20 &   0.32 &   
2.532 &   29.31 &    5.8 & \ref{sect:Bmulti} & \ref{fig:00145a}, &
\ref{fig:00145b} \\
00166 & 2406 & 02 22 22.42 & +57 11 02 & ~ &   13.85 &   0.27 &   0.08 &   
3.707 &   42.90 &    1.9 & \ref{sect:Bmulti} & \ref{fig:00166a}, &
\ref{fig:00166b} \\
00183 & 2324 & 02 22 11.26 & +57 06 33 & A0 &   13.91 &   0.22 &   0.03 &   
3.680 &   42.59 &    4.0 & \ref{sect:Bmulti} & \ref{fig:00183a}, &
\ref{fig:00183b} \\
00196 & 2531 & 02 22 42.97 & +57 12 19 & ~ &   14.10 &   0.26 &  --0.07 &   
3.664 &   42.41 &    3.7 & \ref{sect:Bmulti} & \ref{fig:00196a}, &
\ref{fig:00196b} \\
00202 & 2410 & 02 22 23.22 & +57 10 05 & A0 &   14.13 &   0.25 &  --0.04 &   
3.342 &   38.68 &    3.2 & \ref{sect:Bmulti} & \ref{fig:00202a}, &
\ref{fig:00202b} \\
00218 & 2323 & 02 22 11.12 & +57 06 24 & A0 &   14.24 &   0.26 &  --0.09 &   
3.974 &   46.00 &    2.2 & \ref{sect:Bmulti} & \ref{fig:00218a}, &
\ref{fig:00218b} \\
02230 & 2633 & 02 23 01.00 & +57 07 17 & ~ &    ~ &   ~ &   ~ &   
8.236 &   95.32 &    2.4 & \ref{sect:Bmulti} & \ref{fig:02230a}, &
\ref{fig:02230b} \\
02293 & 2448 & 02 22 29.02 & +57 01 43 & ~ &    ~ &   ~ &   ~ &   
0.497 &    5.75 &    7.4 & \ref{sect:Bmulti} & \ref{fig:02293a}, &
\ref{fig:02293b} \\
02299 & 2649 & 02 23 04.18 & +57 07 39 & Be &    ~ &   ~ &   ~ &   
2.244 &   25.97 &   14.8 & \ref{sect:Bmulti} & \ref{fig:02299a}, &
\ref{fig:02299b} \\
02300 & 2601 & 02 22 56.81 & +57 11 44 & ~ &   10.64 &   0.30 &   0.59 &   
6.702 &   77.57 &    2.4 & \ref{sect:Bmulti} & \ref{fig:02300a}, &
\ref{fig:02300b} \\
02320 & 2089 & 02 21 43.00 & +57 00 54 & ~ &    ~ &   ~ &   ~ &   
0.883 &   10.22 &    5.8 & \ref{sect:Bmulti} & \ref{fig:02320a}, &
\ref{fig:02320b} \\
02430 & 2694 & 02 23 11.62 & +57 10 20 & ~ &    ~ &   ~ &   ~ &   
3.733 &   43.20 &    5.4 & \ref{sect:Bmulti} & \ref{fig:02430a}, &
\ref{fig:02430b} \\
02436 & 2520 & 02 22 41.81 & +57 16 59 & ~ &    ~ &   ~ &   ~ &   
5.657 &   65.48 &    2.7 & \ref{sect:Bmulti} & \ref{fig:02436a}, &
\ref{fig:02436b} \\
02451 & 2086 & 02 21 44.00 & +57 18 21 & ~ &    ~ &   ~ &   ~ &  
25.275 &  292.54 &    2.5 & \ref{sect:Bmulti} & \ref{fig:02451a}, &
\ref{fig:02451b} \\
02540 & 2753 & 02 23 22.00 & +57 00 26 & ~ &    ~ &   ~ &   ~ &   
2.233 &   25.85 &    7.6 & \ref{sect:Bmulti} & \ref{fig:02540a}, &
\ref{fig:02540b} \\
00006 & 2371 & 02 22 17.67 & +57 07 24 & B2III &    9.20 &   0.19 &   0.81 &   
0.384 &    4.44 &   20.2 & \ref{sect:Bmono} & \ref{fig:00006a}, &
\ref{fig:00006b} \\
00008 & 2235 & 02 22 00.92 & +57 08 14 & B1V &    9.37 &   0.19 &   0.80 &   
0.211 &    2.44 &    1.0 & \ref{sect:Bmono} & \ref{fig:00008a}, &
\ref{fig:00008b} \\
00017 & 2566 & 02 22 48.10 & +57 12 01 & B1ne &   10.47 &   0.32 &   0.74 &   
2.552 &   29.54 &   13.3 & \ref{sect:Bmono} & \ref{fig:00017a}, &
\ref{fig:00017b} \\
00018 & 2262 & 02 22 04.55 & +57 10 38 & B2Vn &   10.55 &   0.24 &   0.77 &   
3.072 &   35.56 &    5.7 & \ref{sect:Bmono} & \ref{fig:00018a}, &
\ref{fig:00018b} \\
00029 & 2377 & 02 22 19.20 & +57 11 36 & B &   11.05 &   0.22 &   0.74 &   
1.388 &   16.06 &    1.6 & \ref{sect:Bmono} & \ref{fig:00029a}, &
\ref{fig:00029b} \\
00039 & 2139 & 02 21 49.64 & +57 08 21 & B2V &   11.36 &   0.18 &   0.76 &   
1.477 &   17.10 &    1.7 & \ref{sect:Bmono} & \ref{fig:00039a}, &
\ref{fig:00039b} \\
00042 & 2462 & 02 22 32.76 & +57 08 41 & ~ &   11.49 &   0.18 &   0.65 &   
1.251 &   14.48 &    2.1 & \ref{sect:Bmono} & \ref{fig:00042a}, &
\ref{fig:00042b} \\
00043 & 2372 & 02 22 18.56 & +57 11 50 & ~ &   11.60 &   0.22 &   0.67 &   
0.766 &    8.86 &    3.3 & \ref{sect:Bmono} & \ref{fig:00043a}, &
\ref{fig:00043b} \\
00057 & 2579 & 02 22 50.28 & +57 08 50 & B3 &   11.91 &   0.19 &   0.56 &   
2.282 &   26.41 &   11.0 & \ref{sect:Bmono} & \ref{fig:00057a}, &
\ref{fig:00057b} \\
00068 & 2189 & 02 21 56.70 & +57 12 49 & ~ &   12.29 &   0.27 &   0.47 &   
0.104 &    1.20 &    3.2 & \ref{sect:Bmono} & \ref{fig:00068a}, &
\ref{fig:00068b} \\
00072 & 2352 & 02 22 15.16 & +57 09 40 & B6 &   12.32 &   0.16 &   0.41 &   
0.291 &    3.37 &    1.8 & \ref{sect:Bmono} & \ref{fig:00072a}, &
\ref{fig:00072b} \\
00082 & 2455 & 02 22 30.35 & +57 03 51 & ~ &   12.54 &   0.18 &   0.51 &   
0.639 &    7.40 &   15.1 & \ref{sect:Bmono} & \ref{fig:00082a}, &
\ref{fig:00082b} \\
00086 & 2285 & 02 22 07.23 & +57 12 11 & ~ &   12.64 &   0.24 &   0.56 &   
0.623 &    7.21 &    9.6 & \ref{sect:Bmono} & \ref{fig:00086a}, &
\ref{fig:00086b} \\
00095 & 2006 & 02 21 28.41 & +57 08 34 & ~ &   12.81 &   0.24 &   0.35 &   
2.703 &   31.29 &    2.6 & \ref{sect:Bmono} & \ref{fig:00095a}, &
\ref{fig:00095b} \\
00099 & 2309 & 02 22 09.57 & +57 06 33 & B7 &   12.81 &   0.19 &   0.38 &   
5.741 &   66.44 &    2.1 & \ref{sect:Bmono} & \ref{fig:00099a}, &
\ref{fig:00099b} \\
00104 & 2037 & 02 21 33.75 & +57 03 56 & ~ &   13.09 &   0.27 &   0.21 &   
0.164 &    1.90 &   16.0 & \ref{sect:Bmono} & \ref{fig:00104a}, &
\ref{fig:00104b} \\
00123 & 1980 & 02 21 24.83 & +57 02 25 & ~ &    ~ &   ~ &   ~ &   
6.266 &   72.52 &    3.7 & \ref{sect:Bmono} & \ref{fig:00123a}, &
\ref{fig:00123b} \\
00131 & 2350 & 02 22 14.89 & +57 09 38 & B7 &   13.36 &   0.22 &   0.31 &   
2.632 &   30.46 &    5.1 & \ref{sect:Bmono} & \ref{fig:00131a}, &
\ref{fig:00131b} \\
00139 & 1990 & 02 21 25.71 & +57 04 22 & ~ &   13.52 &   0.29 &   0.12 &   
2.974 &   34.42 &    5.2 & \ref{sect:Bmono} & \ref{fig:00139a}, &
\ref{fig:00139b} \\
00142 & 2116 & 02 21 46.91 & +57 09 54 & B8 &   13.50 &   0.29 &   0.17 &   
3.213 &   37.19 &    1.7 & \ref{sect:Bmono} & \ref{fig:00142a}, &
\ref{fig:00142b} \\
00146 & 2426 & 02 22 25.98 & +57 06 53 & B8V &   13.75 &   0.25 &   0.07 &   
0.752 &    8.70 &    1.5 & \ref{sect:Bmono} & \ref{fig:00146a}, &
\ref{fig:00146b} \\
00162 & 2414 & 02 22 23.84 & +57 08 26 & A2 &   13.83 &   0.22 &  --0.27 &   
0.437 &    5.06 &    7.7 & \ref{sect:Bmono} & \ref{fig:00162a}, &
\ref{fig:00162b} \\
00163 & 2524 & 02 22 41.66 & +57 10 28 & ~ &   13.86 &   0.30 &  --0.02 &   
3.222 &   37.29 &    2.0 & \ref{sect:Bmono} & \ref{fig:00163a}, &
\ref{fig:00163b} \\
\hline
00188 & 2562 & 02 22 47.45 & +57 11 27 & ~ &   14.20 &   0.42 &  --0.14 &   
2.574 &   29.79 &    7.7 & \ref{sect:Bmono} & \ref{fig:00188a}, &
\ref{fig:00188b} \\
00201 & 2370 & 02 22 17.74 & +57 06 13 & B9V &   14.02 &   0.21 &   0.03 &   
3.274 &   37.90 &    4.0 & \ref{sect:Bmono} & \ref{fig:00201a}, &
\ref{fig:00201b} \\
00238 & 2482 & 02 22 35.99 & +57 12 32 & ~ &   14.44 &   0.19 &  --0.20 &   
0.702 &    8.12 &    2.8 & \ref{sect:Bmono} & \ref{fig:00238a}, &
\ref{fig:00238b} \\
02196 & 1973 & 02 21 20.25 & +57 06 27 & ~ &    ~ &   ~ &   ~ &   
4.507 &   52.16 &    1.4 & \ref{sect:Bmono} & \ref{fig:02196a}, &
\ref{fig:02196b} \\
02231 & 2622 & 02 22 59.99 & +57 12 14 & ~ &    ~ &   ~ &   ~ &   
2.081 &   24.08 &    5.5 & \ref{sect:Bmono} & \ref{fig:02231a}, &
\ref{fig:02231b} \\
02233 & 2141 & 02 21 49.63 & +57 01 36 & ~ &    ~ &   ~ &   ~ &   
0.209 &    2.42 &   54.6 & \ref{sect:Bmono} & \ref{fig:02233a}, &
\ref{fig:02233b} \\
02234 & 2616 & 02 22 58.42 & +57 04 49 & ~ &    ~ &   ~ &   ~ &   
3.166 &   36.64 &    3.0 & \ref{sect:Bmono} & \ref{fig:02234a}, &
\ref{fig:02234b} \\
02321 & 2342 & 02 22 13.13 & +57 01 22 & ~ &    ~ &   ~ &   ~ &   
2.678 &   30.99 &    4.5 & \ref{sect:Bmono} & \ref{fig:02321a}, &
\ref{fig:02321b} \\
02351 & 2151 & 02 21 51.69 & +57 15 05 & ~ &    ~ &   ~ &   ~ &   
5.590 &   64.70 &    2.0 & \ref{sect:Bmono} & \ref{fig:02351a}, &
\ref{fig:02351b} \\
02438 & 2563 & 02 22 46.99 & +56 58 06 & B2IIIe &    ~ &   ~ &   ~ &   
3.664 &   42.41 &    5.0 & \ref{sect:Bmono} & \ref{fig:02438a}, &
\ref{fig:02438b} \\
02465 & 2019 & 02 21 31.23 & +57 17 27 & ~ &    ~ &   ~ &   ~ &   
5.109 &   59.14 &    4.1 & \ref{sect:Bmono} & \ref{fig:02465a}, &
\ref{fig:02465b} \\
02468 & 2611 & 02 22 57.42 & +57 00 32 & ~ &    ~ &   ~ &   ~ &   
4.703 &   54.43 &    3.1 & \ref{sect:Bmono} & \ref{fig:02468a}, &
\ref{fig:02468b} \\
02475 & 2752 & 02 23 22.91 & +57 13 11 & ~ &    ~ &   ~ &   ~ &   
5.991 &   69.34 &    2.6 & \ref{sect:Bmono} & \ref{fig:02475a}, &
\ref{fig:02475b} \\
02520 & 2725 & 02 23 18.49 & +57 15 33 & ~ &    ~ &   ~ &   ~ &   
1.811 &   20.96 &    4.6 & \ref{sect:Bmono} & \ref{fig:02520a}, &
\ref{fig:02520b} \\
02542 & 2146 & 02 21 49.95 & +56 58 00 & ~ &    ~ &   ~ &   ~ &   
4.013 &   46.44 &    3.2 & \ref{sect:Bmono} & \ref{fig:02542a}, &
\ref{fig:02542b} \\
02552 & 1898 & 02 21 10.93 & +56 59 32 & ~ &    ~ &   ~ &   ~ &   
3.648 &   42.22 &    3.1 & \ref{sect:Bmono} & \ref{fig:02552a}, &
\ref{fig:02552b} \\
00003 & 2296 & 02 22 07.36 & +57 06 42 & B1III &    8.52 &   0.19 &   0.82 &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:00003}}\\
00007 & 2284 & 02 22 06.43 & +57 05 25 & B2III-IVe &    9.65 &   0.28 &   0.88
&  ~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:00007}}\\
00009 & 2088 & 02 21 43.38 & +57 07 33 & B1IIIe &    9.44 &   0.20 &   0.94 &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:00009}}\\
00015 & 2165 & 02 21 52.92 & +57 09 59 & B2Ve &   10.04 &   0.20 &   0.83 &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:00015}}\\
00037 & 2085 & 02 21 42.92 & +57 05 30 & B3 &   11.22 &   0.25 &   0.74 &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:00037}}\\
00059 & 1977 & 02 21 24.91 & +57 11 53 & B2 &   12.24 &   0.29 &   0.56 &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:00059}}\\
02431 & 2402 & 02 22 22.76 & +57 17 04 & B1III &    ~ &   ~ &   ~ &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:02431}}\\
02434 & 1926 & 02 21 18.09 & +57 18 22 & B1IIIe &    ~ &   ~ &   ~ &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:02434}}\\
02447 & 2759 & 02 23 25.00 & +57 19 04 & B2 &    ~ &   ~ &   ~ &   
~ &    ~ &    ~ & \ref{sect:Bnoper} & \multicolumn{2}{c}{\ref{fig:02447}}\\
00187 & 1970 & 02 21 23.66 & +57 06 30 & ~ &   14.49 &   0.64 &  --0.39 &   
1.888 &   21.85 &    2.7 & \ref{sect:AFmulti} & \ref{fig:00187a}, &
\ref{fig:00187b} \\
00236 & 2500 & 02 22 37.92 & +57 03 11 & ~ &   14.88 &   0.64 &  --0.35 &   
0.229 &    2.65 &   17.5 & \ref{sect:AFmulti} & \ref{fig:00236a}, &
\ref{fig:00236b} \\
00267 & 2544 & 02 22 43.97 & +57 03 54 & ~ &   15.02 &   0.53 &  --0.41 &   
0.941 &   10.89 &    4.4 & \ref{sect:AFmulti} & \ref{fig:00267a}, &
\ref{fig:00267b} \\
00298 & 2430 & 02 22 27.71 & +57 13 32 & ~ &   15.14 &   0.49 &  --0.53 &   
3.904 &   45.19 &    2.6 & \ref{sect:AFmulti} & \ref{fig:00298a}, &
\ref{fig:00298b} \\
00315 & 2306 & 02 22 09.40 & +57 04 21 & ~ &   15.04 &   0.34 &  --0.48 &  
35.337 &  408.99 &    3.3 & \ref{sect:AFmulti} & \ref{fig:00315a}, &
\ref{fig:00315b} \\
00322 & 2243 & 02 22 03.02 & +57 11 19 & ~ &   15.33 &   0.48 &  --0.57 &  
35.128 &  406.58 &    4.5 & \ref{sect:AFmulti} & \ref{fig:00322a}, &
\ref{fig:00322b} \\
00342 & 2416 & 02 22 24.16 & +57 05 48 & A3 &   15.27 &   0.36 &  --0.49 &  
15.298 &  177.06 &    1.6 & \ref{sect:AFmulti} & \ref{fig:00342a}, &
\ref{fig:00342b} \\
00344 & 2454 & 02 22 29.93 & +57 04 34 & ~ &   15.47 &   0.51 &  --0.67 &   
0.840 &    9.73 &    7.6 & \ref{sect:AFmulti} & \ref{fig:00344a}, &
\ref{fig:00344b} \\
00364 & 2081 & 02 21 42.53 & +57 11 17 & ~ &   15.37 &   0.43 &  --0.56 &  
35.754 &  413.82 &    2.4 & \ref{sect:AFmulti} & \ref{fig:00364a}, &
\ref{fig:00364b} \\
00370 & 2461 & 02 22 32.43 & +57 05 49 & ~ &   15.50 &   0.45 &  --0.52 &  
32.249 &  373.25 &    1.6 & \ref{sect:AFmulti} & \ref{fig:00370a}, &
\ref{fig:00370b} \\
00371 & 4990 & 02 21 57.60 & +57 06 16 & ~ &   15.64 &   0.58 &  --0.41 &   
1.083 &   12.53 &   13.6 & \ref{sect:AFmulti} & \ref{fig:00371a}, &
\ref{fig:00371b} \\
00381 & 2476 & 02 22 34.55 & +57 09 47 & ~ &   15.47 &   0.42 &  --0.55 &  
36.614 &  423.78 &    1.8 & \ref{sect:AFmulti} & \ref{fig:00381a}, &
\ref{fig:00381b} \\
00384 & 2220/3379 & 02 21 59.96 & +57 05 10 & ~ &   15.49 &   0.44 &  --0.57 &
 35.236 &  407.82 &    2.8 & \ref{sect:AFmulti} & \ref{fig:00384a}, &
\ref{fig:00384b} \\
00388 & 5155 & 02 22 06.79 & +57 04 29 & ~ &   15.56 &   0.46 &  --0.47 &  
15.584 &  180.37 &    3.6 & \ref{sect:AFmulti} & \ref{fig:00388a}, &
\ref{fig:00388b} \\
00419 & 5168 & 02 21 50.35 & +57 04 36 & ~ &   15.68 &   0.50 &  --0.59 &  
23.346 &  270.21 &    5.5 & \ref{sect:AFmulti} & \ref{fig:00419a}, &
\ref{fig:00419b} \\
00471 & 5151 & 02 22 16.04 & +57 04 51 & ~ &   15.82 &   0.47 &  --0.48 &  
23.650 &  273.73 &    4.7 & \ref{sect:AFmulti} & \ref{fig:00471a}, &
\ref{fig:00471b} \\
00540 & 3276 & 02 22 08.99 & +57 09 55 & ~ &   16.07 &   0.54 &  --0.29 &  
22.962 &  265.76 &    9.6 & \ref{sect:AFmulti} & \ref{fig:00540a}, &
\ref{fig:00540b} \\
00586 & 4830 & 02 22 38.31 & +57 09 03 & ~ &   16.61 &   0.70 &  --0.35 &   
0.728 &    8.43 &    8.6 & \ref{sect:AFmulti} & \ref{fig:00586a}, &
\ref{fig:00586b} \\
02249 & 7057 & 02 22 58.89 & +57 03 45 & ~ &    ~ &   ~ &   ~ &   
8.411 &   97.35 &    3.7 & \ref{sect:AFmulti} & \ref{fig:02249a}, &
\ref{fig:02249b} \\
00332 & 2445 & 02 22 29.55 & +57 09 37 & ~ &   15.18 &   0.35 &  --0.57 &  
42.204 &  488.47 &    1.3 & \ref{sect:AFmono} & \ref{fig:00332a}, &
\ref{fig:00332b} \\
00376 & 5172 & 02 21 52.42 & +57 02 15 & ~ &    ~ &   ~ &   ~ &  
10.585 &  122.51 &    9.1 & \ref{sect:AFmono} & \ref{fig:00376a}, &
\ref{fig:00376b} \\
00389 & 2062 & 02 21 38.00 & +57 02 17 & ~ &    ~ &   ~ &   ~ &  
15.710 &  181.83 &    2.9 & \ref{sect:AFmono} & \ref{fig:00389a}, &
\ref{fig:00389b} \\
00412 & 3344 & 02 21 50.11 & +57 06 49 & ~ &   15.72 &   0.53 &  --0.53 &  
23.169 &  268.16 &    2.3 & \ref{sect:AFmono} & \ref{fig:00412a}, &
\ref{fig:00412b} \\
00417 & 2206 & 02 21 58.56 & +57 07 11 & A7 &   15.58 &   0.43 &  --0.49 &  
23.440 &  271.30 &    4.5 & \ref{sect:AFmono} & \ref{fig:00417a}, &
\ref{fig:00417b} \\
00421 & 2030 & 02 21 32.95 & +57 04 35 & ~ &   15.69 &   0.48 &  --0.53 &  
18.238 &  211.09 &    4.7 & \ref{sect:AFmono} & \ref{fig:00421a}, &
\ref{fig:00421b} \\
00428 & 5127 & 02 22 34.24 & +57 03 11 & ~ &   15.70 &   0.45 &  --0.52 &  
20.088 &  232.50 &    2.0 & \ref{sect:AFmono} & \ref{fig:00428a}, &
\ref{fig:00428b} \\
00438 & ~ & 02 22 34.75 & +57 12 40 & ~ &   15.75 &   0.45 &  --0.55 &  
22.420 &  259.50 &    3.5 & \ref{sect:AFmono} & \ref{fig:00438a}, &
\ref{fig:00438b} \\
00443 & 4832 & 02 22 30.65 & +57 09 36 & ~ &   15.77 &   0.46 &  --0.45 &  
49.457 &  572.42 &    1.9 & \ref{sect:AFmono} & \ref{fig:00443a}, &
\ref{fig:00443b} \\
00450 & 2339 & 02 22 13.12 & +57 07 54 & ~ &   15.76 &   0.44 &  --0.61 &  
48.455 &  560.82 &    2.2 & \ref{sect:AFmono} & \ref{fig:00450a}, &
\ref{fig:00450b} \\
00462 & 4713 & 02 21 36.69 & +57 12 59 & ~ &   15.90 &   0.50 &  --0.53 &  
20.528 &  237.59 &    3.7 & \ref{sect:AFmono} & \ref{fig:00462a}, &
\ref{fig:00462b} \\
00497 & 4829 & 02 22 37.34 & +57 08 19 & ~ &   15.92 &   0.46 &  --0.48 &  
41.166 &  476.46 &    2.1 & \ref{sect:AFmono} & \ref{fig:00497a}, &
\ref{fig:00497b} \\
00503 & 3363 & 02 21 46.70 & +57 06 32 & ~ &   15.96 &   0.52 &  --0.51 &  
22.753 &  263.35 &    2.6 & \ref{sect:AFmono} & \ref{fig:00503a}, &
\ref{fig:00503b} \\
00509 & 3212 & 02 22 11.13 & +57 05 46 & ~ &   15.98 &   0.49 &  --0.48 &  
22.379 &  259.01 &    2.5 & \ref{sect:AFmono} & \ref{fig:00509a}, &
\ref{fig:00509b} \\
00527 & 4819 & 02 22 42.59 & +57 09 54 & ~ &   15.93 &   0.37 &  --0.58 &  
20.696 &  239.54 &    2.1 & \ref{sect:AFmono} & \ref{fig:00527a}, &
\ref{fig:00527b} \\
00570 & 5160 & 02 22 02.62 & +57 03 29 & ~ &   16.31 &   0.55 &  --0.41 &   
1.577 &   18.26 &   22.7 & \ref{sect:AFmono} & \ref{fig:00570a}, &
\ref{fig:00570b} \\
00577 & 5183 & 02 21 36.26 & +57 04 27 & ~ &   16.50 &   0.64 &  --0.45 &   
8.258 &   95.57 &    4.9 & \ref{sect:AFmono} & \ref{fig:00577a}, &
\ref{fig:00577b} \\
00744 & 4978 & 02 22 22.90 & +57 05 04 & ~ &   16.57 &   0.39 &  --0.56 &  
15.022 &  173.87 &    3.7 & \ref{sect:AFmono} & \ref{fig:00744a}, &
\ref{fig:00744b} \\
02205 & 1929 & 02 21 18.49 & +57 09 26 & ~ &    ~ &   ~ &   ~ &  
35.590 &  411.92 &    4.4 & \ref{sect:AFmono} & \ref{fig:02205a}, &
\ref{fig:02205b} \\
02376 & 2210 & 02 21 59.76 & +57 15 25 & ~ &    ~ &   ~ &   ~ &   
9.172 &  106.16 &    5.2 & \ref{sect:AFmono} & \ref{fig:02376a}, &
\ref{fig:02376b} \\
02414 & ~ & 02 21 37.41 & +57 14 12 & ~ &   17.16 &   0.59 &  --0.45 &   
3.910 &   45.26 &   16.3 & \ref{sect:AFmono} & \ref{fig:02414a}, &
\ref{fig:02414b} \\
\hline
02727 & 2671 & 02 23 08.19 & +57 01 02 & ~ &    ~ &   ~ &   ~ &  
35.767 &  413.97 &    3.1 & \ref{sect:AFmono} & \ref{fig:02727a}, &
\ref{fig:02727b} \\
02740 & 2703 & 02 23 14.41 & +57 11 40 & ~ &    ~ &   ~ &   ~ &   
1.569 &   18.16 &   17.8 & \ref{sect:AFmono} & \ref{fig:02740a}, &
\ref{fig:02740b} \\
02833 & 4398 & 02 21 55.01 & +57 18 42 & ~ &    ~ &   ~ &   ~ &  
46.878 &  542.57 &    4.4 & \ref{sect:AFmono} & \ref{fig:02833a}, &
\ref{fig:02833b} \\
00521 & 3267 & 02 22 15.62 & +57 09 45 & ~ &   16.02 &   0.55 &  --0.05 &   
1.980 &   22.91 &   19.0 & \ref{sect:other} & \ref{fig:00521a}, &
\ref{fig:00521b} \\
00565 & 5145 & 02 22 21.64 & +57 04 00 & ~ &   16.71 &   0.91 &  --0.63 &   
2.291 &   26.52 &   13.0 & \ref{sect:other} & \ref{fig:00565a}, &
\ref{fig:00565b} \\
00631 & 6911 & 02 22 26.75 & +57 11 29 & ~ &   16.97 &   0.83 &  --0.72 &   
2.367 &   27.40 &   15.1 & \ref{sect:other} & \ref{fig:00631a}, &
\ref{fig:00631b} \\
00681 & 4839 & 02 22 15.05 & +57 10 58 & ~ &   17.18 &   0.94 &  --0.68 &   
1.818 &   21.04 &   28.6 & \ref{sect:other} & \ref{fig:00681a}, &
\ref{fig:00681b} \\
00719 & ~ & 02 22 12.60 & +57 06 03 & ~ &   17.10 &   0.88 &  --0.42 &   
0.862 &    9.98 &   30.1 & \ref{sect:other} & \ref{fig:00719a}, &
\ref{fig:00719b} \\
00726 & ~ & 02 22 31.28 & +57 04 54 & ~ &   17.26 &   0.89 &  --0.53 &   
2.238 &   25.90 &   17.3 & \ref{sect:other} & \ref{fig:00726a}, &
\ref{fig:00726b} \\
00765 & 2359 & 02 22 15.37 & +57 08 26 & B9 &   17.35 &   0.98 &  --0.60 &   
1.872 &   21.67 &   19.2 & \ref{sect:other} & \ref{fig:00765a}, &
\ref{fig:00765b} \\
00805 & ~ & 02 22 10.49 & +57 03 55 & ~ &   17.41 &   0.50 &  --0.18 &   
2.420 &   28.01 &   18.8 & \ref{sect:other} & \ref{fig:00805a}, &
\ref{fig:00805b} \\
00816 & 6870 & 02 22 19.69 & +57 09 53 & ~ &   17.49 &   0.89 &  --0.60 &   
0.779 &    9.02 &   42.3 & \ref{sect:other} & \ref{fig:00816a}, &
\ref{fig:00816b} \\
00827 & ~ & 02 21 39.01 & +57 11 08 & ~ &   17.48 &   0.88 &  --0.60 &   
0.239 &    2.77 &   30.6 & \ref{sect:other} & \ref{fig:00827a}, &
\ref{fig:00827b} \\
01098 & ~ & 02 21 54.01 & +57 06 31 & ~ &   18.40 &   1.01 &  --0.76 &   
6.682 &   77.34 &   47.2 & \ref{sect:other} & \ref{fig:01098a}, &
\ref{fig:01098b} \\
02334 & 5329 & 02 22 21.68 & +57 01 31 & ~ &    ~ &   ~ &   ~ &   
9.524 &  110.23 &   10.2 & \ref{sect:other} & \ref{fig:02334a}, &
\ref{fig:02334b} \\
02478 & 2681/5292 & 02 23 09.25 & +56 59 48 & ~ &    ~ &   ~ &   ~
&  0.058 &    0.67 &   12.9 & \ref{sect:other} & \ref{fig:02478a}, &
\ref{fig:02478b} \\
02507 & 1964 & 02 21 22.73 & +57 01 24 & ~ &    ~ &   ~ &   ~ &   
6.953 &   80.48 &    8.2 & \ref{sect:other} & \ref{fig:02507a}, &
\ref{fig:02507b} \\
02513 & 2741 & 02 23 19.18 & +56 56 10 & ~ &    ~ &   ~ &   ~ &   
1.181 &   13.67 &   14.6 & \ref{sect:other} & \ref{fig:02513a}, &
\ref{fig:02513b} \\
02534 & 2670/4943 & 02 23 08.62 & +57 07 07 & ~ &    ~ &   ~ &   ~ &
  0.032 &   0.37 &   30.9 & \ref{sect:other} & \ref{fig:02534a}, &
\ref{fig:02534b} \\
02543 & 2779 & 02 23 29.14 & +57 14 11 & ~ &    ~ &  ~ &   ~ &  
29.031 &  336.01 &    2.7 & \ref{sect:other} & \ref{fig:02543a}, &
\ref{fig:02543b} \\
02909 & 1866 & 02 21 06.16 & +57 11 37 & ~ &    ~ &   ~ &   ~ &   
1.591 &   18.41 &   19.2 & \ref{sect:other} & \ref{fig:02909a}, &
\ref{fig:02909b} \\
02953 & 5101 & 02 23 18.10 & +57 02 30 & ~ &    ~ &   ~ &   ~ &   
0.103 &    1.19 &   54.2 & \ref{sect:other} & \ref{fig:02953a}, &
\ref{fig:02953b} \\
03114 & 5205 & 02 21 11.64 & +57 03 59 & ~ &    ~ &   ~ &   ~ &   
2.189 &   25.33 &   27.6 & \ref{sect:other} & \ref{fig:03114a}, &
\ref{fig:03114b} \\
00091 & 2107 & 02 21 45.54 & +57 05 30 & ~ &   13.88 &   1.16 &  --1.29 &   
0.824 &    9.54 &    2.1 & \ref{sect:other} & \ref{fig:00091a}, &
\ref{fig:00091b} \\
00507 & 4947 & 02 22 54.02 & +57 07 34 & ~ &   16.57 &   0.93 &  --0.71 &   
0.331 &    3.83 &   35.6 & \ref{sect:other} & \ref{fig:00507a}, &
\ref{fig:00507b} \\
00549 & 3242 & 02 22 17.91 & +57 06 48 & ~ &   16.70 &   0.96 &  --0.54 &   
2.707 &   31.34 &   11.1 & \ref{sect:other} & \ref{fig:00549a}, &
\ref{fig:00549b} \\
00660 & 3209 & 02 22 16.19 & +57 05 39 & ~ &   17.03 &   0.87 &  --0.61 &   
0.570 &    6.60 &   20.8 & \ref{sect:other} & \ref{fig:00660a}, &
\ref{fig:00660b} \\
00667 & 5176 & 02 21 45.55 & +57 02 25 & ~ &   16.76 &  --0.51 &   0.01 &   
2.113 &   24.46 &   32.0 & \ref{sect:other} & \ref{fig:00667a}, &
\ref{fig:00667b} \\
00706 & ~ & 02 21 43.93 & +57 07 52 & ~ &   17.06 &   0.73 &  --0.11 &   
5.084 &   58.84 &  126.7 & \ref{sect:other} & \ref{fig:00706a}, &
\ref{fig:00706b} \\
00710 & 5175 & 02 21 46.45 & +57 02 44 & ~ &   17.36 &   1.08 &  --0.73 &   
2.594 &   30.02 &   15.5 & \ref{sect:other} & \ref{fig:00710a}, &
\ref{fig:00710b} \\
00738 & ~ & 02 22 17.23 & +57 07 32 & ~ &   16.50 &  --0.07 &   3.19 &   
3.013 &   34.87 &   58.5 & \ref{sect:other} & \ref{fig:00738a}, &
\ref{fig:00738b} \\
00834 & ~ & 02 21 39.77 & +57 11 19 & ~ &   17.80 &   1.02 &  --0.67 &   
8.081 &   93.53 &   12.1 & \ref{sect:other} & \ref{fig:00834a}, &
\ref{fig:00834b} \\
00907 & ~ & 02 22 07.42 & +57 13 42 & ~ &   17.70 &   0.69 &  --0.67 &   
5.478 &   63.41 &   50.3 & \ref{sect:other} & \ref{fig:00907a}, &
\ref{fig:00907b} \\
00940 & 4716 & 02 21 25.34 & +57 11 21 & ~ &   18.05 &   1.06 &  --0.90 &   
3.711 &   42.95 &   53.4 & \ref{sect:other} & \ref{fig:00940a}, &
\ref{fig:00940b} \\
02476 & 2791 & 02 23 30.41 & +57 02 42 & ~ &    ~ &   ~ &   ~ &   
5.645 &   65.33 &    1.9 & \ref{sect:other} & \ref{fig:02476a}, &
\ref{fig:02476b} \\
02480 & 1900 & 02 21 12.21 & +57 12 47 & ~ &    ~ &   ~ &   ~ &   
0.329 &    3.80 &   10.4 & \ref{sect:other} & \ref{fig:02480a}, &
\ref{fig:02480b} \\
02684 & 1915 & 02 21 16.13 & +57 17 28 & ~ &    ~ &   ~ &   ~ &  
24.667 &  285.50 &    6.0 & \ref{sect:other} & \ref{fig:02684a}, &
\ref{fig:02684b} \\
02826 & 4728 & 02 21 00.65 & +57 13 33 & ~ &    ~ &   ~ &   ~ &   
0.409 &    4.74 &   38.9 & \ref{sect:other} & \ref{fig:02826a}, &
\ref{fig:02826b} \\
02860 & 4471 & 02 23 23.05 & +57 14 42 & ~ &    ~ &   ~ &   ~ &  
13.464 &  155.83 &    4.6 & \ref{sect:other} & \ref{fig:02860a}, &
\ref{fig:02860b} \\
02865 & 5226 & 02 21 03.34 & +57 02 54 & ~ &    ~ &   ~ &   ~ &  
35.193 &  407.33 &    4.4 & \ref{sect:other} & \ref{fig:02865a}, &
\ref{fig:02865b} \\
02900 & 4489 & 02 22 53.26 & +57 16 55 & ~ &    ~ &   ~ &   ~ &  
11.339 &  131.24 &    5.8 & \ref{sect:other} & \ref{fig:02900a}, &
\ref{fig:02900b} \\
03122 & 5466 & 02 22 32.16 & +56 58 13 & ~ &    ~ &   ~ &   ~ &  
12.034 &  139.29 &   51.6 & \ref{sect:other} & \ref{fig:03122a}, &
\ref{fig:03122b} \\
03146 & ~ & 02 21 32.62 & +57 00 32 & ~ &    ~ &   ~ &   ~ &   
2.698 &   31.23 &   25.8 & \ref{sect:other} & \ref{fig:03146a}, &
\ref{fig:03146b} \\
00001 & 2417 & 02 22 24.28 & +57 06 34 & M4Iab &   ~ &   ~ &  ~ &   
~ &    ~ &    ~ & \ref{sect:other} & \multicolumn{2}{c}{\ref{fig:00001}}\\
00002 & 2227 & 02 22 00.56 & +57 08 41 & B2Ibp &    ~ &   ~ &   ~ &   
~ &    ~ &    ~ & \ref{sect:other} & \multicolumn{2}{c}{\ref{fig:00002}}\\
02427 & 2691/4637 & 02 23 11.07 & +57 11 58 & M1Iab &    ~ &   ~ &   ~ &
  ~ &    ~ &    ~ & \ref{sect:other} & \multicolumn{2}{c}{\ref{fig:02427}}\\
02428 & 2758/4624 & 02 23 24.13 & +57 12 44 & M1Iab &    ~ &   ~ &   ~ &
  ~ &    ~ &    ~ & \ref{sect:other} & \multicolumn{2}{c}{\ref{fig:02428}}\\
03164 & 1818 & 02 20 59.63 & +57 09 31 & sdM1 &    ~ &   ~ &   ~ &   
~ &    ~ &    ~ & \ref{sect:other} & \multicolumn{2}{c}{\ref{fig:03164}}\\
03165 & 2589 & 02 22 53.51 & +57 14 42 & A2Ia &    ~ &   ~ &  ~ &   
~ &    ~ &    ~ & \ref{sect:other} & \multicolumn{2}{c}{\ref{fig:03165}}\\
\hline
\end{tabular}
\end{table}